Peatland carbon dynamics in distinctively warmer world

Nitin Chaudhary¹, Wenxin Zhang¹, Shubhangi Lamba², and Sebastian Westermann³

¹Lund University ²University of Gothenburg ³University of Oslo

November 22, 2022

Abstract

Peatlands store large amounts of carbon in the terrestrial ecosystem and it seems that they would shift from being a carbon sink to a carbon source due to recent warming. In this study, we provide a comprehensive analysis of peatland carbon dynamics in distinct future climate conditions. The study examined whether less pronounced warming could further enhance the peatland carbon sink capacity and buffer the effects of climate change. It also determined which trajectory peatland carbon balance would follow, what the main drivers were, and which one would dominate in the future. We found that peatlands could largely retain their carbon sink capacity with reduced rates under the climate change scenario RCP 2.6 to RCP 6.0. They are projected to shift from a carbon sink to a carbon-neutral or a small carbon source in the strong warming scenario RCP 8.5 by the end of this century.

Hosted file

essoar.10507599.1.docx available at https://authorea.com/users/534092/articles/598385-peatland-carbon-dynamics-in-distinctively-warmer-world

Peatland carbon dynamics in distinctively warmer world

Nitin Chaudhary $^{1,2},$ Wenxin Zhang $^{1,3},$ Shubhang
i Lamba 4 and Sebastian Westermann 2

^{1.} Department of Physical Geography and Ecosystem Science, Lund University,

Sölvegatan 12, SE- 22362 Lund, Sweden

^{2.} Department of Geosciences, University of Oslo,

Sem Sælands vei 1, Geologibygningen, 0371 Oslo, Norway

^{3.} Department of Geosciences and Natural Resource Management,

University of Copenhagen, Øster Voldgade 10, 1350 Copenhagen, Denmark

^{4.} Department of Biological and Environmental Sciences,

University of Gothenburg, 40530 Gothenburg, Sweden

Abstract

Peatlands store large amounts of carbon in the terrestrial ecosystem and it seems that they would shift from being a carbon sink to a carbon source due to recent warming. In this study, we provide a comprehensive analysis of peatland carbon dynamics in distinct future climate conditions. The study examined whether less pronounced warming could further enhance the peatland carbon sink capacity and buffer the effects of climate change. It also determined which trajectory peatland carbon balance would follow, what the main drivers were, and which one would dominate in the future. We found that peatlands could largely retain their carbon sink capacity with reduced rates under the climate change scenario RCP 2.6 to RCP 6.0. They are projected to shift from a carbon sink to a carbon-neutral or a small carbon source in the strong warming scenario RCP 8.5 by the end of this century.

Plain language summary

Peatlands are important terrestrial ecosystems that take up carbon from the atmosphere but ongoing warming could shift this balance. In this study, we investigated how peatland would behave in the distinct warmer conditions and whether less to moderate warming could increase their carbon sink capacity and reduce the effects of climate change. We found that peatlands would remain a major sink of carbon by the end of this century, although their carbon accumulation rates would somewhat decrease compared to present conditions. Conversely, their sink capacity would substantially reduce in an extremely warmer world and there is a risk that they would turn into a carbon source or sequestering carbon by 2100.

Key Points:

• The effects of climate change on peatland carbon balance under different warming scenarios were determined

- Peatlands will likely remain carbon sink until 2100 in low to moderate warming scenarios
- They are projected to shift from a carbon sink to carbon neutral or a small carbon source in the strong warming scenario RCP 8.5

1. Introduction

Peatlands are considered one of the biggest carbon reserves in the terrestrial ecosystem, comprising 30% of the present-day soil organic carbon pool (Yu 2012). They are also a major source of methane (CH_4) , a potent greenhouse gas (Abdalla et al. 2016). Recent observations have shown that the vegetation structure, hydrology, and carbon balance are rapidly changing in many peatlands (Johansson et al. 2006, Pinceloup et al. 2020). These ongoing changes will disturb the prevailing land-atmosphere carbon balance and trigger some pertinent climate-relevant feedbacks (Belyea 2013, Zhu and Zhuang 2016). Studies have indicated that peatlands will continue to act as a carbon sink in the next decades under different warming scenarios, but they will likely become carbon neutral or even a major carbon source by the end of this century (Chaudhary et al. 2017b, Gallego-Sala et al. 2018, Chaudhary et al. 2020, Qiu et al. 2020). To quantify and understand the overall effects of these rapid changes, various advanced peatland models have been employed, but often these models are forced with a distinct set of RCP scenarios (Chaudhary et al. 2020, Qiu et al. 2020, Müller and Joos 2021). It is a common trend to focus on two-three scenarios in modelling studies in order to obtain end-member estimates. All the climate scenarios are developed as a suite of complementary possibilities and the Intergovernmental Panel of Climate Change (IPCC) recommends that modelling studies should consider all the Representative Concentration Pathways (RCP) scenarios for a complete understanding of system behaviour in future conditions (IPCC 2013a). Furthermore, Ritchie and Dowlatabadi (2017) recommend that the ultra-high warming scenario RCP85 should be used with caution for future scientific research or policy studies because the amount of coal that this scenario requires by major fossil fuel consuming countries assume to use in coming decades will likely exceed the estimates of readily available fossil resources. Experts believe that RCP45 and RCP60 are the more realistic scenarios (IPCC 2013b, Ritchie and Dowlatabadi 2017), yet these two scenarios are often omitted from modelling studies.

The purpose of this study is to provide a comprehensive analysis of peatland carbon dynamics in the future climate conditions branching all plausible future warming scenarios. The study will examine whether less pronounced warming can further enhance the peatland carbon sink capacity and buffer the effects of climate change. It will also determine which trajectory peatland carbon balance will follow, what the main drivers are and which one will dominate in the future. This study is a continuation of our previous study (Chaudhary et al. 2020), where we analyzed long-term carbon accumulation rates in the past and in the future, using only a limited number of model points to simulate the observed peatland area under two warming scenarios. In this study, we progress

a step forward and perform simulations at one-degree resolution within the observed peatland domain delineated by PEATMAP (Xu et al. 2018) to obtain a an overview of disguised responses of peatlands to four different warming scenarios by the end of the current century. Furthermore, we identify and highlight hotspot regions which are vulnerable to additional warming and likely switch its ecosystem functions between carbon sources or sinks according to our simulations.

2. Methodology

2.1 LPJ-GUESS peatland

LPJ-GUESS (Lund-Potsdam-Jena General Ecosystem Simulator; (Smith et al. 2001, Smith et al. 2014) is a second-generation dynamic global vegetation model (DGVM) that is widely employed in global carbon cycle and vegetation dynamics studies (e.g. Saunois et al., 2020). LPJ-GUESS is included as a land surface scheme in the global Earth System Model (ESM) EC-Earth (Alessandri et al. 2017), and the regional ESM, such as RCA-GUESS (Wramneby et al. 2010, Zhang et al. 2014) and RCAO-GUESS (Zhang et al. 2020). LPJ-GUESS simulates vegetation structure and composition in response to a changing climate from local to global scales. It is a process-based model of vegetation dynamics which incorporates physiological changes and biogeochemistry of terrestrial ecosystems. It dynamically simulates water and carbon fluxes through coupled vegetation, soil, and hydrological interaction. Recently, new peatland and permafrost formulations with a unique representation of spatial heterogeneity were implemented in LPJ-GUESS (Chaudhary et al. 2017a). In particular, dynamic annual multi-layer peat accumulation, freezing-thawing cycles, lateral flow, and spatial heterogeneity in the framework of the dynamic vegetation model have for the first time been considered in the model (see Table S1 in Chaudhary et al. (2017a)), with applications from local to regional scales. Our previous studies have demonstrated that the mechanistic multi-layer peat accumulation scheme can simulate vegetation dynamics, permafrost, and peat distribution across the pan-Arctic region in a reasonably robust way. The current scheme consists of many important key mechanisms and interactions controlling the non-linear peatland dynamics (Chaudhary et al. 2018). Phenomena such as the 'humpbacked' relation between the average rate of peat formation and water table position, the cyclicity among micro-formations, internal eco-hydrological feedbacks and multi-directionality, which have been frequently observed in many peatland sites, can be simulated and explained using this detailed model scheme. In this study, we have employed the customized Arctic version of LPJ-GUESS that includes dynamic peat accumulation and decomposition functionalities with a freeze-thaw cycle. In the model, peat accumulates due to an imbalance between annual litter input and decomposition. The decomposition rate is controlled by the litter quality and thermo-hydrological conditions within the peat soil. The peat layers become more recalcitrant when sufficient labile peat material is decomposed over time. The current set up also features a unique individualand patch-based representation of small-scale heterogeneity where the different

vegetation units compete for resources, such as light and space. Over time, some patches gain height, while the surface elevation of others decreases or remains unaffected. The adjustment in the height of these dynamic patches drives the water flow from elevated to low-lying patches and affects vegetation and biochemical properties of peatlands. For more information about the model and its functionalities, refer to Chaudhary et al. (2017a)

Using this model, we have performed four experiments with all RCP warming scenarios: RCP26, RCP45, RCP60, and RCP85. The future climate forcing for these RCP scenarios were obtained from the Coupled Model Intercomparison Project Phase 5 (CMIP5) runs with the IPSL-CM5 (Dufresne et al. 2013). The model is forced with daily temperature, precipitation, and cloudiness data which were constructed by interpolating monthly values. The peat initiation surface was constructed using peat basal age values (Chaudhary et al. 2020) and this surface determines the length of the simulated peat accumulation period. The climate data are divided in three different periods: Holocene climate, transient, and future runs. In the Holocene experiment, the model was run from prescribed basal age in calendar years before present (cal. BP) until the year 1900. In the transient run, the CRU TS 3.0 global gridded climate data set (Mitchell and Jones 2005) was used until the year 2005 and the future runs were performed from 2006 to 2100 using RCP warming scenarios. The model output variables examined for this study include recent and near-future pan-Arctic carbon accumulation rates, net primary productivity, decomposition rates, dominant plant cover and permafrost distribution. A detailed description of model structure, data requirement, and simulation protocol is available in Chaudhary et al. (2020) and references therein.

3. Results and Discussion

3.1 Peatland carbon accumulation rates in the 21st century

Modelled northern peatlands have accumulated carbon at a rate of 20-35 g C m $^{-2}$ vear $^{-1}$ between 1990-2000 (see Figure 1) with an average of 25.3 g C m^{-2} vear⁻¹. It was found that carbon sink capacity of the peatlands sharply declines in the first two decades of the 21st century under all warming experiments, reaching a point beyond which each scenario followed their trajectories regulated by their respective thermo-hydrological regimes and internal dynamics. In the RCP85 experiment, the carbon sink capacity of peatlands is projected to continuously steadily reduce throughout the entire period. This decline in peat carbon accumulation rate is quite sharp in the first and last two decades of the 21st century and peatlands overall become near neutral or even a small net carbon source by the end of this century under RCP85. In the RCP26 and RCP45 experiments, similar declining trends in the carbon sink capacity have been noticed, but at comparatively smaller magnitudes, with average carbon accumulation rates remaining between 10-20 g C m⁻² year⁻¹ after 2020. In the RCP60 experiment, after an initial decline in the first two decades, carbon accumulation rates stabilize and fluctuate between 15-20 g C m⁻² vear⁻¹. largely remaining in the range of average long-term Holocene carbon accumulation rates. The peatland carbon sink capacity declines marginally after 2060 and then strengthens again in the last few decades after 2080 in this scenario. It is interesting to note that there is virtually no difference between RCP26, RCP45 and RCP60, which indicates that peatlands are quite robust to future climate changes for a wide range of future warming trends (Fig. 1). However, there is a clear climatic threshold to RCP85, beyond which the system starts behaving very differently, fully loosing its capacity to sequester carbon by the end of this century.

Overall, it is projected that peatlands turn into a small source of carbon by the end of this century in RCP85, while peatland carbon accumulation rates remain within the range of long-term Holocene carbon accumulation rates under the moderate warming experiment RCP60. In the ultra-low and low warming experiments (RCP26 and RCP45), peatlands remain a major sink of carbon by the end of this century, although the carbon accumulation rates are somewhat decreased compared to today.

If we look at the spatial distribution of peatland carbon accumulation rates, the majority of peatlands showed an increase in their carbon accumulation rates from the 19th to 20th century (Fig. 2b). The mean carbon accumulation rates at the beginning of the 20^{th} century were predicted to be around 25 g C m⁻² year⁻¹ which steadily increased to 30-32 g C m⁻² year⁻¹ by the end of the 20th century. Peatlands located in central and Eastern Europe, northeastern US, southeastern Canada and along the Russian-Chinese border were found to be most vulnerable to additional climate warming and predicted to become a strong source of carbon, as the climate becomes progressively warmer. Conversely, peatlands in Siberia, the Hudson Bay lowlands and western Canada strengthen their carbon sink capacity. In short, low latitude peatlands are predicted to become a greater source of carbon, but other regions enhance their carbon sink capacity and counteract those losses in the model (Fig. 2c-j). The moderate warming since the 19th century benefited the overall peatland carbon accumulation rates, but this gain was rapidly compensated by a warming-driven increase in decomposition rates in the first two decades of this century (Figs. 1) and 3).

It is interesting to note that peatlands have already experienced enhanced warming in the first two decades which means that the majority of them may already have a reduced carbon sink capacity and carbon accumulation rates. If future warming follows comparatively moderate warming trend (RCP60), the carbon accumulation rates of peatlands are projected to recover after an initial decrease, but the non-mitigation high-end scenario RCP85 paints a bleak picture. To better understand the differences between the two scenarios, longer runs beyond 2100 are required to get a clearer picture of whether the rates stabilize or continue to decrease, in which case peatlands could eventually become a strong source of carbon.

3.2 Controls of vegetation productivity and decomposition rates on peat accumulation

The main uncertainties in quantifying the peatland carbon balance in the coming century arise from future trajectories of primary productivity and respiration rates. Some studies note that net primary productivity will override decomposition rates and accelerate the carbon accumulation (Wilson et al. 2017, Zhang et al. 2018a), while others argue that the respiration rates remain high and gains in the net primary productivity do not catch up to temperature-driven microbial decomposition (Hugelius et al. 2020). Our results suggest that the soil respiration will override the increases in net primary productivity in the RCP85 experiment (Fig. 3), while soil respiration and net primary productivity change by more similar magnitudes in the other scenarios, so that the overall carbon accumulation does not change strongly.

The main mechanism behind the higher productivity in the model is a temperature-dependent spring onset of photosynthesis which leads to a longer growing season in a warmer climate. This results in plants allocating additional production to the canopy, leading to increased leaf area and canopy growth, which further intercepts light and increases production. The increase in the modelled primary productivity is similar to the one observed in tundra warming experiments and satellite data (Olsrud et al. 2010, Berner et al. 2020). On the other hand, decomposition is exponentially related to soil temperature in the model, so temperature increases can substantially amplify the respiration rates. From Figs. 3 and 4, it can be seen that the net primary productivity increases as the temperature gets warmer particularly in mid and low latitude regions, but the increase in primary productivity is compensated by an increase in soil respiration rates. The respiration rates are higher in low latitude regions and remain moderate (5-10 g C m⁻²) in mid and high latitudes.

Our results suggest that low latitude areas will be dominated by tall shrubs, while higher latitudes feature mosses, graminoids, and dwarf shrubs, collectively characterized as tundra vegetation. As the climate gets warmer, high latitude plant types will lose their ecological niche in many areas. While areas with high moisture content maintain their plant cover assemblage of mosses and graminoids, some areas show encroachment by tall shrubs. The net primary productivity in these regions is projected to increase, leading to increased carbon accumulation in the soil. The underlying mechanism is an increase in light availability which favours taller shrubs by out-competing smaller ground vegetation through shading. Shrub expansion and densification due to recent warming trends have been documented in many studies (Rundqvist et al. 2011, Liljedahl et al. 2020) which is in line with our findings.

3.3 Permafrost distribution

We found that almost 58% of peatlands were underlined by permafrost in beginning of the 20th century which reduced to 56% by the end of it. Hugelius et al. (2020) also found that that almost half of the peatlands are underlain by permafrost which affects their biogeochemistry and carbon cycle. Permafrost peatlands are characterized by ice-rich conditions and observations show that these ice-rich peatlands have been continuously thawing, which changes net primary productivity and carbon accumulation rates. Our simulation results suggest that today's permafrost peatlands start to thaw under all future warming scenarios, resulting in deeper active layer depths (Fig. 6a and b), while the rate of permafrost loss is dependent on the warming scenario. In line with recent studies (Zhang et al. 2018b, Turetsky et al. 2019, Hugelius et al. 2020), the permafrost extent starts shrinking from the southern limit in our simulations, creating space for new vegetation assemblages, which in turn modifies the carbon accumulation rates in those regions. Permafrost-peatlands are predicted to completely disappear from many regions within a few decades, underlining their vulnerability to ongoing warming. According to our simulations, permafrost peatland fractions are predicted to reduce to 39.91%, 35.36%, 35.39% and 29.1% under RCP26, RCP45, RCP45, RCP6.0 and RCP85 respectively.

Conclusion

Peatlands are an important long-term carbon sink in the global climate system and the ongoing warming trend has the potential to modify the peatland carbon balance by the end of this century. In this study, we used a state-of-the-art peatland model to simulate the peatland carbon balance in mid and high latitude regions of the Northern Hemisphere. We found that peatlands can largely retain their carbon sink capacity for the climate change scenario RCP 2.6 to RCP 6.0, but are projected to shift from a carbon sink to carbon neutral or a small carbon source in the strong warming scenario RCP 8.5. While the past warming in the $19^{\rm th}$ and $20^{\rm th}$ century has increased the carbon accumulation rates across the pan-Arctic, those gains seem to be reverse by further warming.

Figures:



Fig. 1. Modelled carbon accumulation (decadal mean) rates for different RCP scenarios



Fig. 2. Modelled carbon accumulation rates (in g C m^{-2}) across pan-Arctic under different RCP warming scenarios



Fig. 3. (a) Modelled net primary productivity and (b) respiration rates across pan-Arctic under different RCP warming scenarios



Fig. 4. Modelled net primary productivity and peat decomposition rates (in kg C m⁻²) in northern peatland sites for different RCP scenarios



Fig. 5. Modelled dominant plant types in northern peatland sites under different RCP scenarios



Fig. 6. Modelled permafrost distribution (in fraction 0 to 1) and active layer depth (in cm) in northern peatland sites for different RCP scenarios

DATA AVAILABILITY

Data archiving is underway in PANGEA data repository (https://www.pangaea.de/)

FUNDING

Nitin Chaudhary acknowledges funding from the Swedish Research Council FORMAS grant (contract no. 2019-01151), Nunataryuk (EU grant agreement no. 773421) and Crafoord grant (no. 20210996). LPJ-GUESS simulations were enabled by resources provided by the Swedish National Infrastructure for Computing (SNIC) at the Lund University Centre for Scientific and Technical Computing (Lunarc), project no. 2021-2-61 and Linköping University, project no. snic2020/5-563. Wenxin Zhang acknowledges the grants from Swedish Research Council FORMAS 2016-01201 and Swedish National Space Agency 209/19. N.C. acknowledge support from the strategic research areas Modeling the Regional and Global Earth System (MERGE) and Biodiversity and Ecosystem Services in a Changing Climate (BECC) at Lund University. We acknowledge the World Climate Research Programme's Working Group on Coupled Modelling, which is responsible for CMIP, and we thank the climate modelling groups for producing and making available their model output.

Reference: Abdalla, M., A. Hastings, J. Truu, M. Espenberg, U. Mander, and

P. Smith. 2016. Emissions of methane from northern peatlands: a review of management impacts and implications for future management options. Ecology and Evolution 6:7080-7102. Alessandri, A., F. Catalano, M. De Felice, B. Van Den Hurk, F. Doblas Reyes, S. Boussetta, G. Balsamo, and P. A. Miller. 2017. Multi-scale enhancement of climate prediction over land by increasing the model sensitivity to vegetation variability in EC-Earth. Climate Dynamics 49:1215-1237.Belyea, L. R. 2013. Nonlinear Dynamics of Peatlands and Potential Feedbacks on the Climate System. In Carbon Cycling in Northern Peatlands, American Geophysical Union: pp 5-18.Berner, L. T., R. Massey, P. Jantz, B. C. Forbes, M. Macias-Fauria, I. Myers-Smith, T. Kumpula, G. Gauthier, L. Andreu-Hayles, B. V. Gaglioti, P. Burns, P. Zetterberg, R. D'Arrigo, and S. J. Goetz. 2020. Summer warming explains widespread but not uniform greening in the Arctic tundra biome. Nature Communications 11:4621. Chaudhary, N., P. A. Miller, and B. Smith. 2017a. Modelling Holocene peatland dynamics with an individual-based dynamic vegetation model. Biogeosciences 14:2571-2596. Chaudhary, N., P. A. Miller, and B. Smith. 2017b. Modelling past, present and future peatland carbon accumulation across the pan-Arctic region. Biogeosciences 14:4023-4044. Chaudhary, N., P. A. Miller, and B. Smith. 2018. Biotic and Abiotic Drivers of Peatland Growth and Microtopography: A Model Demonstration. Ecosystems 21:1196-1214. Chaudhary, N., S. Westermann, S. Lamba, N. Shurpali, B. K. Sannel, G. Schurgers, P. A. Miller, and B. Smith. 2020. Modelling past and future peatland carbon dynamics across the pan-Arctic. Global Change Biology.Dufresne, J. L., M. A. Foujols, S. Denvil, A. Caubel, O. Marti, O. Aumont, Y. Balkanski, S. Bekki, H. Bellenger, R. Benshila, S. Bony, L. Bopp, P. Braconnot, P. Brockmann, P. Cadule, F. Cheruy, F. Codron, A. Cozic, D. Cugnet, N. de Noblet, J. P. Duvel, C. Ethé, L. Fairhead, T. Fichefet, S. Flavoni, P. Friedlingstein, J. Y. Grandpeix, L. Guez, E. Guilyardi, D. Hauglustaine, F. Hourdin, A. Idelkadi, J. Ghattas, S. Joussaume, M. Kageyama, G. Krinner, S. Labetoulle, A. Lahellec, M. P. Lefebvre, F. Lefevre, C. Levy, Z. X. Li, J. Lloyd, F. Lott, G. Madec, M. Mancip, M. Marchand, S. Masson, Y. Meurdesoif, J. Mignot, I. Musat, S. Parouty, J. Polcher, C. Rio, M. Schulz, D. Swingedouw, S. Szopa, C. Talandier, P. Terray, N. Viovy, and N. Vuichard. 2013. Climate change projections using the IPSL-CM5 Earth System Model: from CMIP3 to CMIP5. Climate Dynamics 40:2123-2165.Gallego-Sala, A. V., D. J. Charman, S. Brewer, S. E. Page, I. C. Prentice, P. Friedlingstein, S. Moreton, M. J. Amesbury, D. W. Beilman, S. Bjorck, T. Blyakharchuk, C. Bochicchio, R. K. Booth, J. Bunbury, P. Camill, D. Carless, R. A. Chimner, M. Clifford, E. Cressey, C. Courtney-Mustaphi, F. De Vleeschouwer, R. de Jong, B. Fialkiewicz-Koziel, S. A. Finkelstein, M. Garneau, E. Githumbi, J. Hribjlan, J. Holmquist, P. D. M. Hughes, C. Jones, M. C. Jones, E. Karofeld, E. S. Klein, U. Kokfelt, A. Korhola, T. Lacourse, G. Le Roux, M. Lamentowicz, D. Large, M. Lavoie, J. Loisel, H. Mackay, G. M. MacDonald, M. Makila, G. Magnan, R. Marchant, K. Marcisz, A. M. Cortizas, C. Massa, P. Mathijssen, D. Mauquoy, T. Mighall, F. J. G. Mitchell, P. Moss, J. Nichols, P. O. Oksanen, L. Orme, M. S. Packalen, S. Robinson, T. P. Roland, N. K. Sanderson, A. B. K. Sannel, N. Silva-Sanchez, N. Steinberg, G. T. Swindles, T. E. Turner, J. Uglow, M. Valiranta, S. van Bellen, M. van der Linden, B. van Geel, G. P. Wang, Z. C. Yu, J. Zaragoza-Castells, and Y. Zhao. 2018. Latitudinal limits to the predicted increase of the peatland carbon sink with warming. Nature Climate Change 8:907. Hugelius, G., J. Loisel, S. Chadburn, R. B. Jackson, M. Jones, G. MacDonald, M. Marushchak, D. Olefeldt, M. Packalen, M. B. Siewert, C. Treat, M. Turetsky, C. Voigt, and Z. Yu. 2020. Large stocks of peatland carbon and nitrogen are vulnerable to permafrost thaw. Proceedings of the National Academy of Sciences 117:20438-20446.IPCC. 2013a. Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.IPCC. 2013b. Summary for Policymakers. Pages 1-30 in T. F. Stocker, D. Qin, G.-K. Plattner, M. Tignor, S. K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex, and P. M. Midgley, editors. Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA. Johansson, T., N. Malmer, P. M. Crill, T. Friborg, J. H. Akerman, M. Mastepanov, and T. R. Christensen. 2006. Decadal vegetation changes in a northern peatland, greenhouse gas fluxes and net radiative forcing. Global Change Biology 12:2352-2369.Liljedahl, A. K., I. Timling, G. V. Frost, and R. P. Daanen. 2020. Arctic riparian shrub expansion indicates a shift from streams gaining water to those that lose flow. Communications Earth & Environment 1:50.Mitchell, T. D., and P. D. Jones. 2005. An improved method of constructing a database of monthly climate observations and associated high-resolution grids. International Journal of Climatology 25:693-712.Müller, J., and F. Joos. 2021. Committed and projected future changes in global peatlands – continued transient model simulations since the Last Glacial Maximum. Biogeosciences 18:3657-3687.Olsrud, M., B. Å. Carlsson, B. M. Svensson, A. Michelsen, and J. M. Melillo. 2010. Responses of fungal root colonization, plant cover and leaf nutrients to long-term exposure to elevated atmospheric CO2 and warming in a subarctic birch forest understory. Global Change Biology 16:1820-1829.Pinceloup, N., M. Poulin, M.-H. Brice, and S. Pellerin. 2020. Vegetation changes in temperate ombrotrophic peatlands over a 35 year period. Plos One 15:e0229146.Qiu, C., D. Zhu, P. Ciais, B. Guenet, and S. Peng. 2020. The role of northern peatlands in the global carbon cycle for the 21st century. Global Ecology and Biogeography 29:956-973.Ritchie, J., and H. Dowlatabadi. 2017. Why do climate change scenarios return to coal? Energy 140:1276-1291.Rundqvist, S., H. Hedenås, A. Sandström, U. Emanuelsson, H. Eriksson, C. Jonasson, and T. V. Callaghan. 2011. Tree and Shrub Expansion Over the Past 34 Years at the Tree-Line Near Abisko, Sweden. Ambio 40:683-692.Saunois, M., et al. 2020. The Global Methane Budget 2000-2017. 2020. Earth Syst. Sci. Data: 12: 1561-1623.Smith, B., I. C. Prentice, and M. T. Sykes. 2001. Representation of vegetation dynamics in the modelling of terrestrial ecosystems: comparing two contrasting approaches within European climate space. Global Ecology and Biogeography 10:621-637.Smith, B., D. Warlind, A. Arneth, T. Hickler, P. Leadley, J. Siltberg, and S. Zaehle. 2014. Implications of incorporating N cycling and N limitations on primary production in an individual-based dynamic vegetation model. Biogeosciences 11:2027-2054.Turetsky, M. R., B. W. Abbott, M. C. Jones, K. W. Anthony, D. Olefeldt, E. A. G. Schuur, C. Koven, A. D. McGuire, G. Grosse, P. Kuhry, G. Hugelius, D. M. Lawrence, C. Gibson, and A. B. K. Sannel. 2019. Permafrost collapse is accelerating carbon release. Nature 569:32-34.Wilson, R. M., L. Fitzhugh, G. J. Whiting, S. Frolking, M. D. Harrison, N. Dimova, W. C. Burnett, and J. P. Chanton. 2017. Greenhouse gas balance over thaw-freeze cycles in discontinuous zone permafrost. Journal of Geophysical Research: Biogeosciences 122:387-404.Wramneby, A., B. Smith, and P. Samuelsson. 2010. Hot spots of vegetation-climate feedbacks under future greenhouse forcing in Europe. Journal of Geophysical Research: Atmospheres 115.Xu, J. R., P. J. Morris, J. G. Liu, and J. Holden. 2018. PEATMAP: Refining estimates of global peatland distribution based on a meta-analysis. Catena 160:134-140.Yu, Z. C. 2012. Northern peatland carbon stocks and dynamics: a review. Biogeosciences 9:4071-4085.Zhang, H., A. V. Gallego-Sala, M. J. Amesbury, D. J. Charman, S. R. Piilo, and M. M. Valiranta. 2018a. Inconsistent Response of Arctic Permafrost Peatland Carbon Accumulation to Warm Climate Phases. Global Biogeochemical Cycles 32:1605-1620.Zhang, H., S. R. Piilo, M. J. Amesbury, D. J. Charman, A. V. Gallego-Sala, and M. M. Valiranta. 2018b. The role of climate change in regulating Arctic permafrost peatland hydrological and vegetation change over the last millennium. Quaternary Science Reviews 182:121-130.Zhang, W., C. Jansson, P. A. Miller, B. Smith, and P. Samuelsson. 2014. Biogeophysical feedbacks enhance the Arctic terrestrial carbon sink in regional Earth system dynamics. Biogeosciences 11:5503-5519.Zhang, W., Döscher, R., Koenigk, T., Miller, P. A., Jansson, C., Samuelsson, P., et al. 2020. The interplay of recent vegetation and sea ice dynamics—Results from a regional Earth system model over the Arctic. Geophysical Research Letters, 47: e2019GL085982.Zhu, X., and Q. Zhuang. 2016. Relative importance between biogeochemical and biogeophysical effects in regulating terrestrial ecosystemclimate feedback in northern high latitudes. Journal of Geophysical Research: Atmospheres 121:5736-5748.